Author's response to reviews

Title: Causes of neonatal and maternal deaths in Dhaka slums: Implications for service delivery

Authors:

Fatema Khatun (kfatema@icddrb.org)  
Sabrina Rasheed (sabrina1@icddrb.org)  
Allisyn C Moran (allisynmoran@gmail.com)  
M Ashraful Alam (aneeloy@yahoo.com)  
Mohammad Sohel Shomik (mshomik@icddrb.org)  
Munira Sultana (sultana.munira16@gmail.com)  
Nuzhat Choudhury (nuzhat@icddrb.org)  
Mohammad Iqbal (migbal@icddrb.org)  
Abbas Bhuiya (abbas@icddrb.org)

Version: 5 Date: 25 January 2012

Author's response to reviews:

Date: January 25, 2012
To: The Editor-in-Chief
BMC Public Health

Subject: Revised manuscript on maternal and neonatal causes of death in urban slums of Dhaka.

Dear Dr. Omar Rahman,

Thank you for giving opportunity for proof reading. In response to the proof reading we have reviewed and revised in minor changes on manuscript entitled “Causes of neonatal and maternal deaths in Dhaka slums: Implications for service delivery” as follows:

1. Deleted some text in analysis part of methods section.
2. Revised reference number 7
3. Revised some part of address on corresponding affiliation (working unit has renamed )

Thank you for your kind consideration.

Yours sincerely,

Dr. Fatema Khatun
Assistant Scientist
Centre for Equity and Health Systems
ICDDR,B 68, Shaheed Tajuddin Ahemd Sarani
Mohakhali, Dhaka-1212
Bangladesh
Fax : 880-2-882 6050,
Email: kfatema@icddrb.org